%&asr
\begin{document}
%Titlepage
\maketitleru[
	course={(Automatic) Speech Recognition},
	institute={Radboud University Nijmegen},
	authorstext={Author:}]
\listoftodos[Todo]

t\cite{muller_multimodal_2012}

t\cite{pedone_phoneme-level_2011}

t\cite{fujihara_automatic_2006}

t\cite{mesaros_adaptation_2009}

t\cite{mesaros_automatic_2010}

t\cite{dzhambazov_automatic_2016}

t\cite{mesaros_automatic_2008}

t\cite{berenzweig_locating_2001}

t\cite{dzhambazov_automatic_2014}

t\cite{fujihara_three_2008}

t\cite{yang_machine_2012}

t\cite{fujihara_lyricsynchronizer:_2011}

t\cite{mauch_integrating_2012}

\bibliographystyle{ieeetr}
\bibliography{asr}
\end{document}
